Snake antivenom for snake venom induced consumption coagulopathy (Review)

Maduwage K, Buckley NA, de Silva HJ, Lalloo DG, Isbister GK



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2015, Issue 6

http://www.thecochranelibrary.com

WILEY

TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
METHODS	3
RESULTS	7
Figure 1	8
DISCUSSION	9
AUTHORS' CONCLUSIONS	9
ACKNOWLEDGEMENTS	10
REFERENCES	10
CHARACTERISTICS OF STUDIES	14
DATA AND ANALYSES	17
APPENDICES	17
CONTRIBUTIONS OF AUTHORS	19
DECLARATIONS OF INTEREST	19
SOURCES OF SUPPORT	19

[Intervention Review]

Snake antivenom for snake venom induced consumption coagulopathy

Kalana Maduwage¹, Nick A Buckley², H Janaka de Silva³, David G Lalloo⁴, Geoffrey K Isbister¹

¹School of Medicine and Public Health, University of Newcastle, Waratah, Australia. ²Department of Pharmacology, University of Sydney, Camperdown, Australia. ³Department of Medicine, University of Kelaniya, Ragama, Sri Lanka. ⁴Clinical Research Group, Liverpool School of Tropical Medicine, Liverpool, UK

Contact address: Kalana Maduwage, School of Medicine and Public Health, University of Newcastle, C/O Calvary Mater Newcastle, Waratah, NSW, 2294, Australia. kalanapm@gmail.com. kalanamaduwage@yahoo.com.

Editorial group: Cochrane Injuries Group. Publication status and date: New, published in Issue 6, 2015. Review content assessed as up-to-date: 30 January 2015.

Citation: Maduwage K, Buckley NA, de Silva HJ, Lalloo DG, Isbister GK. Snake antivenom for snake venom induced consumption coagulopathy. *Cochrane Database of Systematic Reviews* 2015, Issue 6. Art. No.: CD011428. DOI: 10.1002/14651858.CD011428.pub2.

Copyright © 2015 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

Snake venom induced consumption coagulopathy is a major systemic effect of envenoming. Observational studies suggest that antivenom improves outcomes for venom induced consumption coagulopathy in some snakebites and not others. However, the effectiveness of snake antivenom in all cases of venom induced consumption coagulopathy is controversial.

Objectives

To assess the effect of snake antivenom as a treatment for venom induced consumption coagulopathy in people with snake bite.

Search methods

The search was done on 30 January 2015. We searched the Cochrane Injuries Group's Specialised Register, the Cochrane Central Register of Controlled Trials (CENTRAL, *The Cochrane Library*), Ovid MEDLINE(R), Ovid MEDLINE(R) In-Process & Other Non-Indexed Citations, Ovid MEDLINE(R) Daily and Ovid OLDMEDLINE(R), Embase Classic+Embase (OvidSP), three other sources, clinical trials registers, and we also screened reference lists.

Selection criteria

All completed, published or unpublished, randomised, controlled trials with a placebo or no treatment arm, where snake antivenom was administered for venom induced consumption coagulopathy in humans with snake bites.

Data collection and analysis

Two authors reviewed the identified trials and independently applied the selection criteria.

Main results

No studies met the inclusion criteria for this review.

Authors' conclusions

Randomised placebo-controlled trials are required to investigate the effectiveness of snake antivenom for clinically relevant outcomes in patients with venom induced consumption coagulopathy resulting from snake bite. Although ethically difficult, the routine administration of a treatment that has a significant risk of anaphylaxis cannot continue without strong evidence of benefit.

PLAIN LANGUAGE SUMMARY

Snake antivenoms for treating people who have been bitten by a snake, and have developed abnormal blood clotting

Many snake venoms cause coagulopathy in humans. Coagulopathy is a condition in which the person's blood is unable to clot because the venom causes decreased levels of clotting factors. Coagulopathy increases the risk of bleeding. Antivenom is a treatment used to neutralise venom in people who have been bitten by a snake. There is some evidence from observational studies in humans which suggest that snake antivenom is helpful to people who have been bitten by a snake. However, the use of antivenom has some risks, and can cause allergic reactions.

Antivenom is made by injecting venom into either horses, sheep or goats, and then collecting the animal blood and separating out the specific antibodies to the snake venom. The antivenom is put into a person's vein, so that it can mix with the blood in their body.

The authors of this Cochrane review investigated whether there was evidence that antivenom helped people who had been bitten by a snake and had developed coagulopathy. The authors looked for studies where antivenom was used as a treatment for people who developed coagulopathy after a snake bite, regardless of the type of snake.

The type of study eligible for inclusion in the review was the randomised controlled trial, and the control group needed to receive either a placebo or no antivenom. The review authors did not find any trials meeting this criteria, despite searching all the major international medical reference databases. The databases were searched on 30 January 2015.

Since no relevant randomised controlled trials were identifed, this systematic review provides no evidence to help doctors decide if and when to use antivenom for snakebite coagulopathy. The authors say that trials of antivenom are urgently needed so that doctors and patients can fully understand the benefits and risks of antivenom. At the moment doctors make decisions about when to use antivenom based on the results of observational studies, which may not fully describe the effects of antivenom.

BACKGROUND

Description of the condition

Snake envenoming is a major medical problem in tropical areas. The estimated burden of snake bite is approximately 421,000 cases of envenoming with 20,000 fatalities annually, although there may be as many as 1,841,000 envenomings and 94,000 deaths (Kasturiratne 2008).

Venom induced consumption coagulopathy is one of the major clinical manifestations of snake envenoming and may be complicated by fatal haemorrhage (Isbister 2010a). Venom induced consumption coagulopathy has previously been referred to by a number of different terms, including disseminated intravascular coagulation, defibrination syndrome and procoagulant coagulopathy (Isbister 2010b). Venom induced consumption coagulopathy results from the action of snake procoagulant toxins on human coagulation factors causing consumption of these clotting factors leading to multiple factor deficiencies (Isbister 2009a). There are many examples of procoagulant snake toxins that cause venom induced consumption coagulopathy, including prothrombin activators in Echis carinatus, Pseudonaja textilis, Notechis scutatus venoms (Rosing 1992; Joseph 2001; Rosing 2001), factor X activators in Dabois russelii, Bothrops atrox, Cerastes cerastes, Bungarus, Ophiophagus venom (Tans 2001), factor V activators in Bothrops atrox, Naja naja oxiana venom (Rosing 2001), thrombin-like enzymes in Agkistrodon contortrix contortrix venom (Swenson 2005), and plasminogen activators in Trimeresurus stejnegeri venom (Sanchez 2006). Venom induced consumption coagulopathy can result in bleeding if there is trauma, or spontaneous haemorrhage in cases where the venom also contains a haemorrhagin (e.g. E. carinatus).

Major haemorrhage in vital organs, such as intracranial haemorrhage, is the most serious issue and is often fatal.

A number of laboratory clotting times and clotting factor studies are used to diagnose and monitor venom induced consumption coagulopathy, including the prothrombin time/international normalised ratio, the activated partial thromboplastin time, and the 20-minute whole blood clotting test. These play a major role in diagnosis, assessment and treatment of venom induced consumption coagulopathy (Isbister 2010a).

Description of the intervention

Antivenom is the primary treatment for snake envenoming (Lalloo 2003; Isbister 2010c). Antivenoms contain polyclonal antibodies raised against one or more snake venoms. They may contain whole immunoglobulins, but more commonly, pepsin or papain digested fragments of immunoglobulins such as F(ab')2 or Fab. They are made by injecting venom into either horses, sheep or goats, and then collecting blood and separating out the specific antibodies to the snake venom. Intravenously administered antivenom in patients with snake envenoming binds to circulating snake toxins which aims to neutralise or eliminate the toxins and thereby prevent or reverse the clinical effects of envenoming. Monovalent antivenoms are raised against a single snake species, while polyvalent antivenoms are raised against more than one species.

Immediate hypersensitivity reactions to the foreign proteins (immunoglobulins) in snake antivenoms are the major adverse effect of antivenom treatment, including life threatening anaphylaxis (Nuchprayoon 1999; Lalloo 2003; Gawarammana 2004; de Silva 2011; Isbister 2012). Manufacturing protocols and methods of snake antivenoms are different in various regions in the world and the standardisation of snake antivenom production remains problematic.

How the intervention might work

Antibodies in the antivenom bind to the toxic components in snake venom. Early administration of antivenom will bind the circulating procoagulant snake toxins and potentially prevent, delay or lessen the severity of venom induced consumption coagulopathy. In the majority of patients who have already developed venom induced consumption coagulopathy, antivenom is used to neutralise circulating toxins and allow recovery of venom induced consumption coagulopathy. However, the effectiveness and mechanism of action of antivenom in already developed venom induced consumption coagulopathy remains unclear (Isbister 2009b). Recovery of the coagulopathy depends on re-synthesising clotting factors which is not directly affected by antivenom administration (Isbister 2010a).

Why it is important to do this review

Even though snake antivenom is the mainstay of the treatment for snake envenoming, there is controversy regarding the effectiveness of antivenom for venom induced consumption coagulopathy (Isbister 2010a). It is unlikely that antivenom can be administered early enough to prevent venom induced consumption coagulopathy because the procoagulant toxins in snake venoms act rapidly (Isbister 2010a). The more important question is whether the administration of antivenom will speed the recovery of venom induced consumption coagulopathy by inactivating the active toxins to allow re-synthesis of clotting factors (Isbister 2010a). Thus only if further factor consumption is occurring due to significant amounts of circulating pro-coagulant venoms, would antivenom be expected to speed recovery.

Recent observational clinical studies on Australian elapid envenoming indicated that neither early (versus late) antivenom nor higher doses of antivenom (> one vial) were associated with more rapid recovery in venom induced consumption coagulopathy (Allen 2009; Isbister 2009b). In contrast, in *Echis* envenoming in Africa, the use of antivenom does appear to speed the recovery of the coagulopathy (Mion 2013). We aim to examine the clinical trial evidence regarding effectiveness of snake antivenom for venom induced consumption coagulopathy from all snake species.

OBJECTIVES

To assess the effects of antivenom for the recovery from venom induced consumption coagulopathy in people with snake envenoming.

METHODS

Criteria for considering studies for this review

Types of studies

All randomised controlled trials (RCTs) in humans.

Types of participants

People of any age with snake envenoming who have already developed snake venom induced consumption coagulopathy. Diagnosis of venom induced consumption coagulopathy must be based on abnormal results from the 20-minute whole blood clotting test or an elevated international normalised ratio of >2.

Types of interventions

Intravenous administration of snake antivenom regardless of the type of antivenom or the dose. People who were not treated with antivenom were the comparison group.

Types of outcome measures

Primary outcomes

• Mortality

Secondary outcomes

· Major haemorrhages

• Time to improve clotting studies (e.g. time to international normalised ratio <2; time to improve 20-minute whole blood clotting test)

- Immediate systemic hypersensitivity reactions
- Serum sickness

Calculation of information size requirements

Snakebite mortality is very variable and has contributors other than venom induced consumption coagulopathy such as neurotoxicity, myotoxicity and acute renal injury. However, for simplicity we have taken the mortality rate from Kasturiratne 2008, which estimates an overall case-fatality of around 5%. Using G*Power (http://www.gpower.hhu.de/en.html), the estimated sample size required in order to show this rate could be halved would require 2504 people in total.

Search methods for identification of studies

In order to reduce publication and retrieval bias we did not restrict our search by language, date or publication status.

Electronic searches

The Cochrane Injuries Group's Trials Search Co-ordinator searched the following:

1. Cochrane Injuries Group Specialised Register (30/01/2015);

2. The Cochrane Central Register of Controlled Trials (*The Cochrane Library*) (issue 1 of 12, 2015);

3. Ovid MEDLINE(R), Ovid MEDLINE(R) In-Process & Other Non-Indexed Citations, Ovid MEDLINE(R) Daily and Ovid OLDMEDLINE(R) 1946 to 30/01/2015;

Embase Classic + Embase (OvidSP) 1947 to 30/01/2015;
 ISI Web of Science: Science Citation Index Expanded

(1970 to 30/01/2015);

6. ISI Web of Science: Conference Proceedings Citation Index-Science (1990 to 30/01/2015);

 Toxicology Literature Online (TOXLINE) (http:// toxnet.nlm.nih.gov/cgi-bin/sis/htmlgen?TOXLINE) (30/01/ 2015);

 8. ClinicalTrials.gov (https://clinicaltrials.gov/) (30/01/2015);
 9. WHO International Clinical Trials Registry Platform (ICTRP) (http://apps.who.int/trialsearch/) (30/01/2015);
 10. OpenGrey (http://www.opengrey.eu/) (30/01/2015).
 We adapted the MEDLINE search strategy illustrated in Appendix 1 as necessary for each of the other databases. We also added search filters, and a modified version of the 'Cochrane Highly Sensitive Search Strategy, for identifying randomised trials in MEDLINE and Embase' (Lefebvre 2011).

Searching other resources

We searched the reference lists of all relevant studies and contacted experts in the field in order to identify ongoing and completed studies. We also ran a search on Google and Google Scholar restricting the search results from 1947 to present, and reviewed over 500 results to find relevant studies (Appendix 1).

Data collection and analysis

We performed this systematic review according to the instructions given in the *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011), and our protocol (Maduwage 2014).

Selection of studies

Two authors (KM and GI) independently screened the titles and abstracts of all articles identified by the search strategy. When either or both authors identified the article as possibly being a report that meets the inclusion criteria, we obtained the full text version of the published article. Both authors reviewed the full text of each article to determine if the article meets the inclusion criteria. There were no disagreements between the two authors about the inclusion of studies. We provided details of the included and excluded studies in the appropriate tables within the review.

The two authors independently reviewed each article that met the inclusion criteria, and extracted data from the article onto a standard data extraction form. We then compared these data forms, which were consistent with each other.

Data extraction and management

Two authors (KM and GI) extracted data on the following items onto a standard form.

• General information about the article (title of the article,

source, publication year, years the study was conducted, language of publication, etc.).

• Clinical trial characteristics: design, diagnostic ascertainment, standard care provided, randomisation, allocation concealment, interventions, drop-out and lost to follow up rates, definitions of outcomes, and methods of outcome assessment.

• Patients: inclusion and exclusion criteria, sample size, baseline characteristics (e.g. age of the patients, past history of bleeding, anticoagulant therapy or coagulation disorders, clinical severity on enrolment, etc.).

• Interventions: type of antivenom (polyvalent or monovalent), manufacturer, dose of antivenom (number of vials or mg), duration of administration, timing of administration of antivenom after the bite.

• Outcomes: mortality, major haemorrhage (according to the definition by the International Society on Thrombosis and Haemostasis), time to improved clotting function defined as either the time to international normalised ratio <2 or time until a negative result of the 20-minute whole blood clotting test, length of hospital stay, systemic hypersensitivity reactions.

Assessment of risk of bias in included studies

In the future if studies are included in this review, two authors (KM and GI) will independently assess the included studies for risk of bias in the following areas. We will assess risk of bias using the suggested domains and guidance provided in the Cochrane Collaboration's tool for assessing risk of bias (Higgins 2011). We will assess random sequence generation (selection bias), allocation concealment (selection bias), blinding of participants and personnel (performance bias), blinding of outcome assessment (detection bias), incomplete outcome data (attribution bias), selective reporting (reporting bias), and other sources of bias (in particular, funding source). If there is insufficient information we will initially judge domains as "unclear risk" and will attempt to clarify the risk of bias by contacting the study authors. We plan to include all studies irrespective of the risk of bias; however, we plan to perform a sensitivity analysis. If the sensitivity analysis shows substantial differences, we will present alternative estimates that exclude studies with high or unclear risk of bias.

Sequence generation of the randomisation process

• Low risk: using random number tables, computer random number generation, coin tossing, stratified or block randomisation, shuffling cards or envelopes, throwing dice, drawing lots or other valid methods

• High risk: "quasi" randomisation, date of birth, day of visit, identification number or record results, alternate allocation

• Unclear risk: not described or not enough information to make a clear judgment

Allocation concealment

• Low risk: allocation concealment is described and would not allow either the investigator or participants to know or influence treatment group assignment at the time of study entry

• Acceptable methods include central randomisation (phone, web, pharmacy) or sequentially numbered, opaque sealed envelopes

• High risk: the method of allocation is not concealed (e.g. random sequence known to staff in advance, envelopes or packaging without all safeguards or a non-randomised and predictable sequence)

• Unclear risk: trial either did not describe the method of allocation concealment or reported an approach that clearly was not adequate

Blinding of participants and personnel

• Low risk: blinding, and unlikely that the blinding could have been broken, or no blinding, or incomplete blinding but outcome unlikely to be influenced

• High risk: no blinding, incomplete or broken blinding and outcome likely to be influenced

• Unclear risk: not described or not enough information to make a clear judgment

Blinding of outcome assessment

• Low risk: blinding of outcome assessors was clearly maintained, or no blinding but measurements unlikely to be influenced

• High risk: no blinding, or broken blinding, and measurements likely to be influenced

• Unclear risk: not described or not enough information to make a clear judgment

Intention-to-treat analysis

• Low risk: specifically reported that intention-to-treat analysis was undertaken by the authors, or report that makes it unmistakable that intention-to-treat was undertaken for the primary analysis

• High risk: no report of an intention-to-treat analysis being conducted

• Unclear risk: not described or not enough information to make a clear judgment

Incomplete outcome data

• Low risk: no missing data, reasons for missing data not related to outcomes, missing data balanced across groups and proportion missing or plausible effect size not enough to have clinically relevant effects

• High risk: reasons related to outcome and imbalance in number or reasons, proportions missing or plausible effect size

enough to have clinically relevant effect, "as treated" analysis with substantial departure from allocation, inappropriate use of imputation

• Unclear risk: not described or not enough information to make a clear judgment

Selectiveness of outcome reporting

• Low risk: method is available and all pre-specified outcomes of interest are reported in the pre-specified way, protocol not available but it is clear that all pre-specified and expected outcomes of interest are reported

• High risk: outcomes not reported as pre-specified or expected e.g. missing, added, subset, unexpected measurement or methods. Outcomes reported are incomplete and cannot enter a meta-analysis

• Unclear risk: not described or not enough information to make a clear judgment

Reporting bias

We will interpret our results cautiously and with an awareness of the likelihood of reporting bias. We will consider using funnel plots.

Other sources of bias

• Low risk: studies appear to be free of other sources of bias such as imprecision (e.g. small sample size), diversity (e.g. inadequate dose, unusual population)

• High risk: baseline imbalance, non-randomised studies, recruitment bias in cluster-randomised trials, inadequate power and/or implausible sample size calculation, early stopping of trial (based on interim analysis of efficacy)

• Unclear risk: not described or not enough information to make a clear judgment

Measures of treatment effect

Dichotomous data

We planned to present dichotomous data outcomes as risk ratios (RRs) with 95% confidence intervals (CIs) for individual trials.

Continuous data

We planned to present continuous data outcomes with mean differences (MDs) and 95% CIs. We planned to calculate the mean difference if possible as these results are easier for clinicians and readers to interpret; and use standardised mean differences (SMDs) when different scales are used in the trials.

Ordinal data

We planned to report the types of adverse events and complications.

Unit of analysis issues

Individual participants are the unit of analysis. To answer our primary question (does antivenom improve venom induced consumption coagulopathy compared to no antivenom treatment) we planned to initially simply combine all active intervention groups of the study into a single group and compare their outcomes to the control groups(s) not receiving antivenom. We may also explore comparison of doses or types of antivenom (post-hoc).

Dealing with missing data

In the future if studies are included in this review, we will contact the authors of the original studies if essential data are missing from their trial reports. If we receive no reply after eight weeks, we will extract the available data from the published reports. We will assess the missing data and attrition rates for each of the included studies and report the number of participants who are included in the final analysis as a proportion of all participants in the study.

Assessment of heterogeneity

In the future if studies are included in this review, we will evaluate statistical heterogeneity using the Chi² test to assess for heterogeneity between trials, and the I² statistic for quantifying heterogeneity across studies (roughly interpreted as follows: 0 to 30%: probably not important; 31 to 60%: may represent moderate heterogeneity; 61 to 75%: may represent substantial heterogeneity; 76 to 100%: very considerable heterogeneity) as outlined in the *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011). We expect considerable heterogeneity due to considerable variation across trials in setting, snake, intervention and outcomes. We intend to use a random-effects model to account for this heterogeneity in any summary estimates of effect. We may also (posthoc) look for plausible explanations of heterogeneity. We will discuss the implications of heterogeneity and how they relate to external validity in the discussion.

Assessment of reporting biases

Systematic difference between reported and unreported findings are referred to as reporting bias. We will include selective outcome reporting assessment as part of the 'Risk of bias table' and also under 'Intention-to-treat analysis'.

We will assess publication biases by using funnel plots when there are at least 10 studies included in the meta-analysis.

Data synthesis

In the future if studies are included in this review, we will analyse the data using the Cochrane Collaboration statistical software Review Manager. We will express results for dichotomous outcomes as RRs with 95% CIs and continuous outcomes as MDs. We will present data in a 'Summary of findings' table according to Grading of Recommendations Assessment, Development and Evaluation (GRADE) guidelines as well as the method described in the *Cochrane Handbook for Systematic Reviews of Interventionss* (Higgins 2011). The table will include mortality, major haemorrhages, time to improved clotting (e.g. time to international normalised ratio <2 or time to normalised result of the 20-min whole blood clotting test), immediate systemic hypersensitivity reactions and serum sickness as outcomes.

We planned to present dichotomous outcomes such as mortality, number of haemorrhages, number of immediate type hypersensitivity reactions, and number of cases of serum sickness as RRs with 95% CIs for individual trials. For dichotomous data meta-analysis we planned to use a Mantel-Haenszel random-effects model. For continuous outcomes (e.g. time to improve clotting studies) that have been recorded as MDs, SMDs or standard deviations (SDs) with 95% CIs, we planned to use an inverse variance randomeffects model. If we were to find two or more studies assessing the same outcomes we will perform meta-analysis. If meta-analysis is not possible we will write a narrative summary of the study findings and follow alternative methods as described in the *Cochrane Handbook for Systematic Reviews of Interventions* (Higgins 2011).

Subgroup analysis and investigation of heterogeneity

Where possible (if sufficient data and information are available) we will perform subgroup analysis based on the following factors, which are thought to affect outcomes after venom induced consumption coagulopathy:

- 1. type of snake envenoming (elapids and viperids);
- 2. type of snake antivenoms;
- 3. dose of antivenom.

Sensitivity analysis

In the future if studies are included in this review, we will restrict sensitivity analyses to include studies with both adequate allocation concealment and blinded outcome assessment.

RESULTS

Description of studies

We did not find any studies for inclusion in this review. There is one ongoing trial (NCT01864200).

Results of the search

The search retrieved 7530 records and after duplicates were removed we screened 5973 records (Figure 1). The search identified one ongoing study, and the results will be included in the review when they become available.



Figure I. Study flow diagram.

Included studies

There are no studies included in this review.

Excluded studies

We excluded 34 of 35 studies after reviewing the full text report. See Characteristics of excluded studies. One ongoing study was identified.

Risk of bias in included studies

There are no studies included in this review.

Effects of interventions

There are no studies included in this review.

DISCUSSION

Summary of main results

We were unable to identify any placebo randomised controlled trials of snake antivenom for venom induced consumption coagulopathy meeting the inclusion criteria. We identified one ongoing trial. There were 32 published and two ongoing studies comparing two or more different antivenoms or comparing different doses of antivenoms for venom induced consumption coagulopathy. Few non-randomised trials including comparison groups without antivenom showed that antivenom was effective for envenoming by some snakes (e.g. *Echis* species in Africa), but not others (e.g. Australasian elapids) (Isbister 2010a; Mion 2013).

Overall completeness and applicability of evidence

There is a lack of evidence to support or refute a benefit of antivenom for venom induced consumption coagulopathy.

Quality of the evidence

There was no evidence to assess the quality of the evidence.

Potential biases in the review process

There are no studies included in this review.

Agreements and disagreements with other studies or reviews

The results of this review agree with the conclusion of a systematic review article that also examined RCTs comparing the effects of different antivenoms (n=14) and different doses of antivenom (n= 5) (Maduwage 2014a). If antivenom was always highly effective in shortening the duration of VICC it might be expected that differences would also be seen commonly in such trials. In nine of 14 studies, the authors concluded equal effectiveness (or ineffectiveness) of two or three antivenoms, and four of five studies investigating different doses or dosing regimens concluded equal effectiveness. Even the six RCTs that concluded a difference between doses or antivenoms, lacked a strong statistical basis for this conclusion (i.e. the difference was not based on intention to treat analysis; three of these trials were very small and lacked the statistical power to support the significance of the differences observed (Warrell 1974; Warrell 1980; Dart 2001), and the others did a post-hoc analysis (a selected time point, sub-group, outcome, or statistical technique that had not been pre-specified) (Ariaratnam 2001; Smalligan 2007; Abubakar 2010).

AUTHORS' CONCLUSIONS

Implications for practice

There are no completed placebo randomised controlled trials of antivenom for venom induced consumption coagulopathy and therefore nothing from this systematic review provides evidence to help clinicians in deciding to use antivenom for venom induced consumption coagulopathy. The effectiveness of administration of antivenom for venom induced consumption coagulopathy will continue to be based on observational studies until placebo randomised controlled trials are undertaken.

Implications for research

Significant mortality and morbidity is associated with snake envenoming (Kasturiratne 2008) so effective treatments are desperately required. Antivenom was introduced for the treatment of snake envenoming over a century ago and its clinical use has been based on in vitro and in vivo animal studies of efficacy, small observational studies and clinical experience. As confirmed in this review there has never been a placebo randomised controlled trial

to demonstrate clinical effectiveness for venom induced consumption coagulopathy. This raises some difficult clinical questions regarding the use of a treatment known to have significant adverse effects (e.g. severe anaphylaxis; Nuchprayoon 1999) where there is no good evidence demonstrating benefit.

Undertaking placebo randomised controlled trials of snake antivenom is a challenge to clinical research and regarded as potentially highly unethical by many clinicians and experts (Gerardo 2014). Such a suggestion would be regarded by some as similar to doing a placebo controlled trial of insulin for diabetes mellitus, such is the overwhelming belief in the benefit of antivenom therapy. Therein lies the inescapable ethical dilemma. How do we undertake the appropriate placebo controlled trial of antivenom to demonstrate clinical effectiveness, if it is regarded as unethical to not give antivenom to some patients.

Well designed observational studies have demonstrated that for some snakes there appears to be a clear benefit of antivenom, speeding the recovery of coagulopathy in *Echis ocellatus* envenoming (Mion 2013), but for others there is little or no benefit, such as Australasian elapids (Isbister 2009b). There is substantial *in vitro* and *in vivo* evidence that antivenom binds toxins and that antivenom can neutralise the procoagulant and anticoagulant effects of venoms (Isbister 2009a). However, it is essential to translate pre-clinical efficacy studies into clinical effectiveness studies (Isbister 2010c), and understand that antivenom may be beneficial for some snake and some clinical syndromes, but not others.

There are a number of precedents where placebo randomised controlled trials have been commenced or completed for different antivenoms. The ongoing study identified in this review is a good example of such a trial. Details of the ethical considerations for this trial have been published (Gerardo 2014). A recently published placebo randomised controlled trial of antivenom for redback spider bite showed no benefit despite decades of belief that it was effective (Isbister 2014). Again, this study required sufficient evidence to justify ethically undertaking a study with a placebo. In envenoming that causes coagulopathy, the safety of this approach could be ensured by first performing observational studies that demonstrate the time to recovery of clotting factors is not strongly influenced by time to antivenom or dose for a particular snake.

The way forward for developing evidence for antivenom treatment in venom induced consumption coagulopathy will be to use novel study designs to introduce placebo arms. For example, undertaking a placebo controlled trial of early antivenom, where all patients will get antivenom at some stage. Such studies will be challenging but are essential to providing sufficient evidence of benefit for a treatment with severe adverse reactions.

A parallel way forward for developing evidence for antivenom treatment in venom induced consumption coagulopathy might be to examine the use of early (on arrival) versus delayed (after blood test results are returned) antivenom. Such studies will be challenging, not least because of the considerable heterogeneity of clinical features of envenoming by a particular species, but are essential to providing evidence that there is a benefit for this treatment that outweighs the considerable risk of severe adverse reactions.

ACKNOWLEDGEMENTS

We would like to acknowledge the Cochrane Injuries Group editors for their advice and support.

This project was supported by the UK National Institute for Health Research, through Cochrane Infrastructure funding to the Cochrane Injuries Group. The views and opinions expressed are those of the authors and do not necessarily reflect those of the Systematic Reviews Programme, NIHR, NHS or the Department of Health.

REFERENCES

References to studies excluded from this review

Abubakar 2010 {published data only}

Abubakar IS, Abubakar SB, Habib AG, Nasidi A, Durfa N, Yusuf PO, Larnyang S, et al. Randomised Controlled Double-Blind Non-Inferiority Trial of Two Antivenoms for Saw-Scaled or Carpet Viper (*Echis ocellatus*) Envenoming in Nigeria. *PLoS Neglected Tropical Diseases* 2010;4(7):e767.

Ariaratnam 2001 {published data only}

Ariaratnam CA, Sjostrom L, Raziek Z, Kularatne SA, Arachchi RW, Sheriff HMR, Theakston RDG, et al. An open, randomized comparative trial of two antivenoms for the treatment of envenoming by Sri Lankan Russell's viper (Daboia russelii russelii). Transactions of the Royal Society of Tropical Medicine and Hygiene 2001;**95**:74–80.

Boyer 2013 {published data only}

Boyer LV, Chase PB, Degan JA, Figge G, Buelna-Romero A, Luchetti C, Alagon A. Subacute coagulopathy in a randomized, comparative trial of Fab and F(ab')2 antivenoms. *Toxicon* 2013;74:101–8.

Bush 2014 {published data only}

Bush SP, Ruha AM, Seifert SA, Morgan DL, Lewis BJ, Arnold TC, Clark RF, et al. Comparison of F(ab')2 versus Fab antivenom for pit viper envenomation: a prospective, blinded, multicenter, randomized clinical trial. *Clinical Toxicology* 2015;**53**:37–45.

Cardoso 1993 {published data only}

Cardoso JL, Fan HW, França FO, Jorge MT, Leite RP, Nishioka SA, Avila A, et al. Randomized comparative trial of three antivenoms in the treatment of envenoming by lance-headed vipers (*Bothrops jararaca*) in Sao Paulo, Brazil. *Quarterly Journal of Medicine* 1993;**86**:315–25.

Cherian 1998 {published data only}

Cherian AM, Jeyaseelan L. A randomized controlled trial to compare the effectiveness of lower dose versus conventional dose of snake antivenom, in cases of snake bite with systemic envenomation. *Journal of Clinical Epidemiology* 1998;**51**(1): 6S.

Dart 2001 {published data only}

Dart RC, Seifert SA, Boyer LV, Clark RF, Hall E, McKinney P, McNally J, et al. A randomized multicenter trial of crotalinae polyvalent immune fab (ovine) antivenom for the treatment for crotaline snakebite in the United States. *Archives of Internal Medicine* 2001;**161**(16):2030–6.

Isbister 2013 {published data only}

Isbister GK, Buckley NA, Page CB, Scorgie FE, Lincz LF, Seldon M, Brown SG, et al. A randomized controlled trial of fresh frozen plasma for treating venom-induced consumption coagulopathy in cases of Australian snakebite (ASP-18). *Journal of Thrombosis and Haemostasis* 2013;**11** (7):1310–8.

Jorge 1995 {published data only}

Jorge MT, Cardoso JLC, Castro SCB, Ribeiro L, Franca FOS, Sbrogio de Almeida ME, Kamiguti AS, et al. A randomized 'blinded' comparison of two doses of antivenom in the treatment of Bothrops envenoming in Sao Paulo, Brazil. *Transactions of the Royal Society of Tropical Medicine* and Hygiene 1995;**89**(1):111–4.

Karnchanachetanee 1994 {published data only}

Karnchanachetanee C, Hanvivatvong O, Mahasandana S. Monospecific antivenin therapy in Russell's viper bite. *Journal of the Medical Association of Thailand* 1994;77(6): 293–7.

Kothari 2001 {published data only}

Kothari D, Bomb BS, Bolya YK, Srivastava S, Verma YN, Vajpeyee A, Chouhan NS, et al. A comparative study of low dose continuous infusion versus high dose intermittent bolus of antisnake venom in snake bite in udaipur zone. APICON 2001. 2001:Abstract number 86.

Meyer 1997 {published data only}

Meyer WP, Habib AG, Onayade AA, Yakubu A, Smith DC, Nasidi A, Daudu IJ, et al. First clinical experiences with a new ovine fab *Echis ocellatus* snake bite antivenom in Nigeria: Randomized comparative trial with Institute Pasteur Serum (Ipser) Africa antivenom. *American Journal of Tropical Medicine and Hygiene* 1997;**56**(3):291–300.

Myint-Lwin 1989 {published data only}

Myint-Lwin, Tin-Nu-Swe, Myint-Aye-Mu, Than-Than, Thein-Than, Tun-Pe. Heparin therapy in Russell's viper bite victims with impending DIC (a controlled trial). *Southeast Asian Journal of Tropical Medicine and Public Health* 1989; **20**(2):271–7.

NCT00639951 {published data only}

NCT00639951. Multicentric, Randomized, Controlled and Comparative Study to Evaluate the Efficacy of Two Treatment Schemes With Antivipmyn® for the Treatment of Snake Bite Envenomation. http://clinicaltrials.gov/show/ NCT00639951 April 2008.

NCT00868309 {published data only}

NCT00868309. A Comparison of Crotalinae (Pit Viper) Equine Immune F(ab)2 Antivenom (Anavip) and Crotalidae Polyvalent Immune Fab, Ovine Antivenom (CroFab) in the Treatment of Pit Viper Envenomation. https:// www.clinicaltrials.gov/ct2/show/results/NCT00868309 April 2009.

Otero 1996 {published data only}

Otero R, Gutierrez JM, Nunez V, Robles A, Estrada R, Segura E, et al. A randomized double-blind clinical trial of two antivenoms in patients bitten by Bothrops atrox in Colombia. The Regional Group on Antivenom Therapy Research (REGATHER). *Transactions of the Royal Society of Tropical Medicine and Hygiene* 1996;**90**(6):696–700.

Otero 1999 {published data only}

Otero R, Gutierrez JM, Rojas G, Nunez V, Diaz A, Miranda E, Uribe AF, et al. A randomized blinded clinical trial of two antivenoms, prepared by caprylic acid or ammonium sulphate fractionation of IgG, in Bothrops and Porthidium snake bites in Colombia: Correlation between safety and biochemical characteristics of antivenoms. *Toxicon* 1999;**37** (6):895–908.

Otero 2006 {published data only}

Otero R, Leon G, Gutierrez JM, Rojas G, Toro MF, Barona J, Rodriguez V, et al. Efficacy and safety of two whole IgG polyvalent antivenoms, refined by caprylic acid fractionation with or without beta-propiolactone, in the treatment of Bothrops asper bites in Colombia. *Transactions* of the Royal Society of Tropical Medicine and Hygiene 2006; **100**(12):1173–82.

Otero-Patino 1998 {published data only}

Otero-Patino R, Cardoso JLC, Higashi HG, Nunez V, Diaz A, Toro MF, Garcia ME, et al. A randomized, blinded, comparative trial of one pepsin-digested and two whole IgG antivenoms for Bothrops snake bites in Uraba, Colombia. The Regional Group on Antivenom Therapy Research (REGATHER). *American Journal of Tropical Medicine and Hygiene* 1998;**58**(2):183–9.

Otero-Patino 2012 {published data only}

Otero-Patino R, Segura A, Herrera M, Angulo Y, Leon G, Gutierrez JM, Barona J, et al. Comparative study of the efficacy and safety of two polyvalent, caprylic acid fractionated [IgG and F(ab')2] antivenoms, in Bothrops asper bites in Colombia. *Toxicon* 2012;**59**(2):344–55.

Pardal 2004 {published data only}

de Oliveira Pardal PP, Souza SM, da Costa Monteiro MRC, Fan HW, Cardoso JLC, Franca FOS, Tomy SC, et al. Clinical trial of two antivenoms for the treatment of Bothrops and Lachesis bites in the north eastern Amazon

region of Brazil. *Transactions of the Royal Society of Tropical Medicine and Hygiene* 2004;**98**(1):28–42.

Paul 2003 {published data only}

Paul V, Prahlad KA, Earali J, Francis S, Lewis F. Trial of heparin in viper bites. *Journal of the Association of Physicians of India* 2003;**51**:163–6.

Paul 2004 {published data only}

Paul V, Pratibha S, Prahlad KA, Earali J, Francis S, Lewis F. High-dose anti-snake venom versus low-dose anti-snake venom in the treatment of poisonous snake bites--a critical study. *Journal of the Association of Physicians of India* 2004; **52**:14–7.

Paul 2007 {published data only}

Paul V, Pudoor A, Earali J, John B, Anil Kumar CS, Anthony T. Trial of low molecular weight heparin in the treatment of viper bites. *Journal of the Association of Physicians of India* 2007;**55**(May):338–42.

Sellahewa 1994 {published data only}

Sellahewa KH, Kumararatne MP, Dassanayake PB, Wijesundera A. Intravenous immunoglobulin in the treatment of snake bite envenoming: a pilot study. *Ceylon Medical Journal* 1994;**39**(4):173–5.

Shah 1986 {published data only}

Shah PK, Chittora MD, Shekhawat JS, Khangaroot D, Vyas MM. Role of heparin in the management of snake (*Echis carinatus*) bite cases. *Journal of the Association of Physicians of India* 1986;**34**(9):621–3.

Smalligan 2004 {published data only}

Smalligan R, Cole J, Brito N, Laing GD, Mertz BL, Manock S, Maudlin J, et al. Crotaline snake bite in the Ecuadorian Amazon: randomised double blind comparative trial of three South American polyspecific antivenoms. *British Medical Journal* 2007;**329**(7475):1129–33.

Srimannarayana 2004 {published data only}

Srimannarayana J, Dutta TK, Sahai A, Badrinath S. Rational use of anti-snake venom (ASV): trial of various regimens in hemotoxic snake envenomation. *Journal of the Association of Physicians of India* 2004;**52**:788–93.

Tariang 1999 {published data only}

Tariang DD, Philip PJ, Alexander G, Macaden S, Jeyaseelan L, Peter JV, Cherian AM. Randomized controlled trial on the effective dose of anti-snake venom in cases of snake bite with systemic envenomation. *Journal of the Association of Physicians of India* 1999;4:369–71.

Thomas 1985 {published data only}

Thomas PP, Jacob J. Randomised trial of antivenom in snake envenomation with prolonged clotting time. *British Medical Journal* 1985;**291**(6489):177–8.

Warrell 1974 {published data only}

Warrell DA, Davidson NM, Omerod LD, Pope HM, Watkins BJ, Greenwood BM, Ried HA. Bites by the sawscaled or carpet viper (*Echis carinatus*): trial of two specific antivenoms. *British Medical Journal* 1974;4(5942):437–40.

Warrell 1976 {published data only}

Warrell DA, Pope HM, Prentice CR. Disseminated intravascular coagulation caused by the carpet viper (*Echis carinatus*): trial of heparin. *British Journal of Haematology* 1976;**33**(3):335–42.

Warrell 1980 {published data only}

Warrell DA, Warrell MJ, Edgar W, Prentice CR, Mathison J, Mathison J. Comparison of Pasteur and Behringwerke antivenoms in envenoming by the carpet viper (*Echis carinatus*). *British Medical Journal* 1980;**280**(6214):607–9.

Warrell 1986 {published data only}

Warrell DA, Looareesuwan S, Theakston RD, Phillips RE, Chanthavanich P, Viravan C, Supanaranond W, et al. Randomized comparative trial of three monospecific antivenoms for bites by the Malayan pit viper (*Calloselasma rhodostoma*) in Southern Thailand: Clinical and laboratory correlations. *American Journal of Tropical Medicine and Hygiene* 1986;**35**(6):1235–47.

References to ongoing studies

NCT01864200 {published data only}

A Randomized, Double-Blind, Placebo-Controlled Study Comparing CroFab® Versus Placebo With Rescue Treatment for Copperhead Snake Envenomation (Copperhead RCT). Ongoing study July 2013.

Additional references

Allen 2009

Allen GE, Brown SGA, Buckly NA, O'Leary MA, Page CB, Currie BJ, et al. Clinical effects and antivenom dosing in brown snake (Pseudonaja spp.) envenoming - Australian snakebite project (ASP-14). *PLoS One* 2009;7(12):e53188.

de Silva 2011

de Silva HA, Pathmeswaran A, Ranasinha CD, Jayamanne S, Samarakoon SB, Hittharage A, et al. Low-dose adrenaline, promethazine, and hydrocortisone in the prevention of acute adverse reactions to antivenom following snakebite: A randomised, double-blind, placebo-controlled trial. *PLoS Medicine* 2011;**8**(5):e1000435.

Gawarammana 2004

Gawarammana IB, Kularatne SA, Dissanayake WP, Kumarasiri RP, Senanayake N, Ariyasena H. Parallel infusion of hydrocortisone +/- chlorpheniramine bolus injection to prevent acute adverse reactions to antivenom for snakebites. *Medical Journal of Australia* 2004;**180**:20–3.

Gerardo 2014

Gerardo CJ, Lavonas EJ, McKinney RE. Ethical considerations in design of a study to evaluate a US Food and Drug Administration-approved indication: antivenom versus placebo for copperhead envenomation. *Clinical Trials* 2014;**11**(5):560–4.

Higgins 2011

Higgins JPT, Green S (editors). *Cochrane Handbook for Systematic Reviews of Interventions* Version 5.1.0 [updated

March 2011]. The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

Isbister 2009a

Isbister GK. Procoagulant snake toxins: laboratory studies, diagnosis and understanding snakebite coagulopathy. *Seminar of Thrombosis and Hemostasis* 2009;**35**:93–103.

Isbister 2009b

Isbister GK, Duffull SB, Brown SG. Failure of antivenom to improve recovery in Australian snakebite coagulopathy. *Quarterly Journal of Medicine* 2009;**102**:563–8.

Isbister 2010a

Isbister GK, Scorgie FE, O'Leary MA, Seldon M, Brown SGA, Lincz LF. Factor deficiencies in venom-induced consumption coagulopathy resulting from Australian elapid envenomation: Australian snakebite project (ASP-10). *Journal of Thrombosis and Haemostasis* 2010;**8**:2504–13.

Isbister 2010b

Isbister GK. Snake bite doesn't cause disseminated intravascular coagulation: coagulopathy and thrombotic microangiopathy in snake envenoming. *Seminar of Thrombosis and Hemostasis* 2010;**36**:444–51.

Isbister 2010c

Isbister GK. Antivenom efficacy or effectiveness: the Australian experience. *Toxicology* 2010;**268**(3):148–54.

Isbister 2012

Isbister GK, Shahmy S, Mohamed F, Abeysinghe C, Karunathilake H, Ariaratnam A. A randomised controlled trial of two infusion rates to decrease reactions to antivenom. *PLoS One* 2012;7(6):e38739.

Isbister 2014

Isbister GK, Page CB, Buckley NA, Fatovich DM, Pascu O, MacDonald SP, Calver LA, et al. Randomized controlled trial of intravenous antivenom versus placebo for latrodectism: the second Redback Antivenom Evaluation (RAVE-II) study. *Annals of Emergency Medicine* 2014;**64**: 620–8.

Joseph 2001

Joseph JS, Kini RM. Snake venom prothrombin activators homologous to blood coagulation factor Xa. *Haemostasis* 2001;**31**:234–40.

Kasturiratne 2008

Kasturiratne A, Wickremasinghe AR, de Silva N, Gunawardena NK, Pathmeswaran A, Premaratna R, et al. The global burden of snakebite: A literature analysis and modelling based on regional estimates of envenoming and deaths. *PLoS Medicine* 2008;**5**(11):e218.

Lalloo 2003

Lalloo D, Theakston RDG. Snake antivenoms. *Journal of Toxicology-Clinical Toxicology* 2003;**41**:277–90.

Lefebvre 2011

Lefebvre C, Manheimer E, Glanville J. Chapter 6: Searching for studies. In: Higgins JPT, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0 (updated March 2011). The Cochrane Collaboration, 2011. Available from www.cochrane-handbook.org.

Maduwage 2014a

Maduwage K, Isbister GK. Current Treatment for Venom-Induced Consumption Coagulopathy Resulting from Snakebite. *PLoS Neglected Tropical Diseases* 2014;**8**(10): e3220.

Mion 2013

Mion G, Larréché S, Benois A, Petitjeans F, Puidupin M. Hemostasis dynamics during coagulopathy resulting from Echis envenomation. *Toxicon* 2013;**76**:103–9.

Nuchprayoon 1999

Nuchprayoon I, Garner P. Interventions for preventing reactions to snake antivenom. *Cochrane Database of Systematic Reviews* 1999, Issue 4. [DOI: 10.1002/ 14651858.CD002153]

Review Manager

The Nordic Cochrane Centre. Review Manager (RevMan). 5.3. Copenhagen: The Cochrane Collaboration, 2014.

Rosing 1992

Rosing J, Tans G. Structural and functional properties of snake venom prothrombin activators. *Toxicon* 1992;**30**: 1515–27.

Rosing 2001

Rosing J, Govers-Riemslag JW, Yukelson L, Tans G. Factor V activation and inactivation by venom proteases. *Haemostasis* 2001;**31**:241–6.

Sanchez 2006

Sanchez EF, Felicori LF, Chavez-Olortegui C, Magalhaes HB, Hermogenes AL, Diniz MRV, et al. Biochemical characterization and molecular cloning of a plasminogen activator proteinase (LV-PA) from bushmaster snake venom. *Biochimica et Biophysica Acta* 2006;**1760**:1762-71.

Smalligan 2007

Smalligan R, Cole J, Brito N, Laing GD, Mertz BL, Manock S, Maudlin J, et al. Crotaline snake bite in the Ecuadorian Amazon: randomised double blind comparative trial of three South American polyspecific antivenoms. *British Medical Journal* 2007;**329**(7475):1129–33.

Swenson 2005

Swenson S, Markland FS Jr. Snake venom fibrin(ogen)olytic enzymes. *Toxicon* 2005;**45**:1021–39.

Tans 2001

Tans G, Rosing J. Snake venom activators of factor X: an overview. *Haemostasis* 2001;**31**:225–33.

References to other published versions of this review

Maduwage 2014

Maduwage K, Buckley NA, de Silva HJ, Lalloo DG, Isbister G. Snake antivenom for snake venom induced consumption coagulopathy. *Cochrane Database of Systematic Reviews* 2014, Issue 12. [DOI: 10.1002/14651858.CD011428]

* Indicates the major publication for the study

CHARACTERISTICS OF STUDIES

Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Abubakar 2010	No placebo control group. Compared two different antivenoms.
Ariaratnam 2001	No placebo control group. Compared two different antivenoms.
Boyer 2013	No placebo control group. Compared two different antivenoms.
Bush 2014	No placebo control group. Compared two different antivenoms.
Cardoso 1993	No placebo control group. Compared three different antivenoms
Cherian 1998	No placebo control group. Compared two different doses of antivenom
Dart 2001	No placebo control group. Compared two different doses of antivenom
Isbister 2013	No placebo control group. Compared antivenom versus antivenom with fresh frozen plasma
Jorge 1995	No placebo control group. Compared two different doses of antivenom
Karnchanachetanee 1994	No placebo control group. Compared two different doses of antivenom
Kothari 2001	No placebo control group. Compared two different doses of antivenom
Meyer 1997	No placebo control group. Compared two different antivenoms.
Myint-Lwin 1989	No placebo control group. Compared antivenom versus antivenom with heparin
NCT00639951	No placebo control group. Compared two different doses of antivenom. Ongoing study
NCT00868309	No placebo control group. Compared two different antivenoms. Ongoing study
Otero 1996	No placebo control group. Compared two different antivenoms.
Otero 1999	No placebo control group. Compared two different antivenoms.
Otero 2006	No placebo control group. Compared two different antivenoms.
Otero-Patino 1998	No placebo control group. Compared three different antivenoms
Otero-Patino 2012	No placebo control group. Compared two different antivenoms.
Pardal 2004	No placebo control group. Compared two different antivenoms.

(Continued)

Paul 2003	No placebo control group. Compared antivenom versus antivenom with heparin
Paul 2004	No placebo control group. Compared two different doses of antivenom
Paul 2007	No placebo control group. Compared antivenom versus antivenom with heparin
Sellahewa 1994	No placebo control group. Compared antivenom versus antivenom with intravenous immunoglobulin
Shah 1986	No placebo control group. Compared antivenom versus antivenom with heparin
Smalligan 2004	No placebo control group. Compared three different antivenoms
Srimannarayana 2004	No placebo control group. Compared three different doses of antivenom
Tariang 1999	No placebo control group. Compared two different doses of antivenom
Thomas 1985	No placebo control group. Compared two different doses of antivenom
Warrell 1974	No placebo control group. Compared two different antivenoms.
Warrell 1976	No placebo control group. Compared antivenom versus antivenom with heparin
Warrell 1980	No placebo control group. Compared two different antivenoms.
Warrell 1986	No placebo control group. Compared two different doses of antivenom

Characteristics of ongoing studies [ordered by study ID]

NCT01864200

Trial name or title	A Randomized, Double-Blind, Placebo-Controlled Study Comparing CroFab® Versus Placebo With Rescue Treatment for Copperhead Snake Envenomation (Copperhead RCT)
Methods	A Randomized, Double-Blind, Placebo-Controlled Study
Participants	 Inclusion Criteria: Envenomation by a copperhead snake. A snake identified by one of the following means: i. Snake or photograph of snake brought to Emergency Department; ii. Patient chooses copperhead from an array of snake photographs; iii. Patient envenomated in an area where only copperheads are endemic; iv. Patient envenomated by a captive copperhead snake Completion of informed consent and eligibility confirmation within 24 hours of envenomation Envenomation on only one extremity, distal to the elbow or knee Clinical evidence of mild or moderate venom effect (limb swelling and/or tenderness) is present (Venom effects need not be progressing.) Patient willing and able to complete follow-up schedule of assessments Patient is able to read, comprehend and sign the IRB approved consent document(s)

Snake antivenom for snake venom induced consumption coagulopathy (Review)

	 Patient is able to read and comprehend the written assessment tools (e.g. DASH, SF-36, etc.) Patient is ≥14 years of age Patient is sober, competent, and able to complete verbal and written informed consent Exclusion Criteria: Patient has clinical evidence of severe venom effect as defined by meeting any one of the following parameters: i. Swelling to an entire extremity (all major joints affected). Lower extremity: i. swelling crossing hip joint. Upper extremity: swelling crossing shoulder joint; ii. INR > 2.0; iii. Platelets <50,000 cells/µL; iv. Fibrinogen <50 mg/dL Compartment syndrome; vi. Systolic Blood Pressure <90 mmHg; vii. More than minimal bleeding; viii. Investigator's clinical discretion Patient has already received antivenom for the management of the current envenomation Patient is pregnant or breastfeeding Patient has a distracting injury or condition with acute pain or functional impairment, and/or is unable to make a reliable self-report of functionality status based solely on the condition of interest Patient had a previous snake envenomation to any body area in the 30 days prior to screening/ enrolment, regardless of whether antivenom was administered for the previous envenomation Patient had an acute traumatic event, surgery, an acute medical event, or exacerbation of a pre-existing medical or surgical condition affecting the envenomated extremity within the 30 days prior to screening/ enrolment Patient has participated in a clinical study involving an investigational pharmaceutical product or device within the 3 months prior to screening that may have impact on clinical outcomes of snakebite Patient has a known history of hypersensitivity to any components of CroFab®, or to papaya or papain Patient has a known history of hypersensitivity to any components of CroFab®, or to papaya or papain
Interventions	Crotilidae polyvalent immune fab (ovine) and placebo
Outcomes	Patient Specific Functional Scale at Day 14
Starting date	July 2013
Contact information	Anna Temu: anna.temu@btgplc.com
Notes	This study completed in March 2015. Its results will be included in this review when published

DATA AND ANALYSES

This review has no analyses.

APPENDICES

Appendix I. Search strategies

Cochrane Injuries Group Specialised Register & Cochrane Central Register of Controlled Trials (The Cochrane Library) #1MESH DESCRIPTOR Snakes #2MESH DESCRIPTOR Boidae #3MESH DESCRIPTOR colubridae EXPLODE ALL TREES #4MESH DESCRIPTOR elapidae EXPLODE ALL TREES #5MESH DESCRIPTOR Viperidae EXPLODE ALL TREES #6(snake* or viper*):TI,AB,KY #7MESH DESCRIPTOR Snake Venoms EXPLODE ALL TREES #8MESH DESCRIPTOR Elapid Venoms EXPLODE ALL TREES #9MESH DESCRIPTOR Viper Venoms EXPLODE ALL TREES #10(snake venom*):TI,AB,KY #11((venom* or bite*) adj3 snake*):TI,AB,KY #12((envenomation or venom-induced or antivenom* or antivenin*)):TI,AB,KY #13(snake adj3 poisonous):TI,AB,KY #14MESH DESCRIPTOR Antivenins EXPLODE ALL TREES #15#1 OR #2 OR #3 OR #4 OR #5 OR #6 OR #7 OR #8 OR #9 OR #10 OR #11 OR #12 OR #13 OR #14 #16* NOT INMEDLINE NOT INEMBASE #17#15 AND #16 Ovid MEDLINE(R), Ovid MEDLINE(R) In-Process & Other Non-Indexed Citations, Ovid MEDLINE(R) Daily and Ovid OLDMEDLINE(R) 1. exp snakes/ or exp boidae/ or exp colubridae/ or exp elapidae/ or exp viperidae/ 2. (snake* or viper*).ab,ti. 3. exp snake venoms/ or exp elapid venoms/ 4. snake venoms.mp. 5. ((venom* or bite*) adj3 snake*).ab,ti. 6. (envenomation or venom-induced or antivenom* or antivenin*).ab,ti. 7. (snake adj3 poisonous).ab,ti. 8. exp Antivenins/ 9. exp Viper Venoms/ 10. 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8 or 9 11. randomi?ed.ab,ti. 12. randomized controlled trial.pt. 13. controlled clinical trial.pt. 14. placebo.ab. 15. clinical trials as topic.sh. 16. randomly.ab. 17. trial.ti. 18. Comparative Study/ 19. 11 or 12 or 13 or 14 or 15 or 16 or 17 or 18 20. (animals not (humans and animals)).sh. 21. 19 not 20

22. 10 and 21 Embase Classic + Embase (OvidSP) 1. exp snakes/ or exp boidae/ or exp colubridae/ or exp elapidae/ or exp viperidae/ 2. (snake* or viper*).ab,ti. 3. exp snake venoms/ or exp elapid venoms/ 4. snake venoms.mp. 5. ((venom* or bite*) adj3 snake*).ab,ti. 6. (envenomation or venom-induced or antivenom* or antivenin*).ab,ti. 7. (snake adj3 poisonous).ab,ti. 8. exp Antivenins/ 9. exp Viper Venoms/ 10. 1 or 2 or 3 or 4 or 5 or 6 or 7 or 8 or 9 11. exp Randomized Controlled Trial/ 12. exp controlled clinical trial/ 13. exp controlled study/ 14. comparative study/ 15. randomi?ed.ab,ti. 16. placebo.ab. 17. *Clinical Trial/ 18. exp major clinical study/ 19. randomly.ab. 20. (trial or study).ti. 21. 11 or 12 or 13 or 15 or 16 or 17 or 18 or 19 or 20 22. exp animal/ not (exp human/ and exp animal/) 23. 21 not 22 24. 10 and 23 ISI Web of Science: Science Citation Index Expanded & Conference Proceedings Citation Index-Science #11#10 AND #7 AND #6 #10#9 OR #8 #9TS=(envenomation or venom-induced or antivenom* or antivenin*) #8TS=(venom* or bite* or poisonous or poison) #7TS=(snake* OR boidae OR colubridae OR elapidae OR viperidae OR viper*) #6#5 AND #4 #5TS=(human*) #4#3 OR #2 OR #1 #3TS=((singl* OR doubl* OR trebl* OR tripl*) SAME (blind* OR mask*)) #2TS=(controlled clinical trial OR controlled trial OR clinical trial OR placebo) #1TS=(randomised OR randomized OR randomly OR random order OR random sequence OR random allocation OR randomly allocated OR at random OR randomized controlled trial) Toxnet (toxicology data network) antivenom* OR antivenin* OR venom-induced OR envenomation OpenGrey (antivenom* OR antivenin* OR venom-induced OR envenomation) AND (poison* OR bite* OR venom* OR venom-induced) WHO International Clinical Trials Registry Platform (ICTRP)

Title:(poison* OR bite* OR venom* OR venom-induced) AND Condition:(antivenom* OR antivenin* OR venom-induced OR envenomation) AND Recruitment status: ALL

Clinicaltrials.gov

(antivenom* OR antivenin* OR venom-induced OR envenomation) AND (poison* OR bite* OR venom* OR venom-induced) Google and Google Scholar

snake envenoming OR snake envenomation OR coagulopathy OR venom induced OR coagulation abnormalities OR snake antivenom OR snake antivenin AND (randomized controlled trial or randomised controlled trial).

CONTRIBUTIONS OF AUTHORS

All authors contributed to this protocol.

DECLARATIONS OF INTEREST

All authors: none known.

SOURCES OF SUPPORT

Internal sources

• Library services, University of Newcastle, NSW, Australia. Support to find the references for the review.

External sources

• No sources of support supplied